Title: Evaluation of the acceptability of Peer Physical Examination (PPE) in medical and osteopathic students: a cross sectional survey

Authors:

    Fabrizio Consorti (fabrizio.consorti@uniroma1.it)
    Rosaria Mancuso (rosaria_mancuso1987@libero.it)
    Annalisa Piccolo (annalisa.piccolo@yahoo.it)
    Giacomo Consorti (giacomo.consorti@gmail.com)
    Joseph Zurlo (joseph.zurlo@gmail.com)

Version: 4 Date: 9 July 2013

Author's response to reviews:

Only one of the reviewers suggested some minor changes.

More in details:

- we corrected the way references were reported in the text
- we changed "inner" to "internal"

The whole text was also checked again and some language corrections were done